Author's response to reviews

Title: The MATISE study: a randomised trial of group art therapy for people with schizophrenia.

Authors:

Mike J Crawford Dr (m.crawford@imperial.ac.uk)
Helen Killaspy Dr (h.killaspy@medsch.ucl.ac.uk)
Eleftheria Kalaitzaki Ms (erk@gprf.mrc.ac.uk)
Barbara Barrett Dr (barbara.m.barrett@kcl.ac.uk)
Sarah Byford Dr (sarah.1.byford@kcl.ac.uk)
Sue Patterson Dr (sue.patterson@imperial.ac.uk)
Tony Soteriou Dr (Tony.Soteriou@awp.nhs.uk)
Tony O&8217;Neal Dr (tony.oneill@qub.ac.uk)
Katie Clayton Ms (Katie.Clayton@candi.nhs.uk)
Anna Maratos Ms (anna.maratos@nhs.net)
Thomas R Barnes Professor (t.r.barnes@imperial.ac.uk)
David Osborn Dr (dosborn@medsch.ucl.ac.uk)
Tony Johnson Dr (tony.johnson@mrc-bsu.cam.ac.uk)
Michael King Professor (m.king@medsch.ucl.ac.uk)
Peter Tyrer Professor (p.tyrer@imperial.ac.uk)
Diane Waller Professor (diane.waller@virgin.net)

Version: 2 Date: 16 August 2010

Author's response to reviews:

1. Author list: the list of authors in the manuscript have been written exactly as they are in the submission system, both in style and order i.e. 'First name Initial Last name'.

2. The title has been written in sentence case with capitalisation only the first word and the study acronym.

3. The Word count has been deleted.

4. All visible vertical lines have been removed from the tables

5. A complete list of all authors of every referenced paper has been included.

6. The paper has been checked for typography errors.